Author's response to reviews

Title: The effect of long term combined yoga practice on the basal metabolic rate of healthy Adults

Authors:

M S Chaya (chayapu@hotmail.com)
Kurpad A V (a.kurpad@iphcr.res.in)
Nagendra H R (hrnagendra@rediffmail.com)
Nagarathna R (rn44@rediffmail.com)

Version: 6 Date: 23 August 2006

Author's response to reviews:

BMC editorial production team,
Dear Editor,
We thank you very much for the email and as per the instructions we are submitting the final corrected version of my manuscript.
The acknowledgement section is not written as it is a institutional self financed study of swami vivekananda yoga research foundation. Kindly let us know if we have to mention it.
We are equally delighted that the manuscript is accepted and is being published.

Thanking you
Chaya

Corresponding author.